Dear editor

Thank you very much for your letter and suggestions on our manuscript. We have modified it as required. Our response to the problems in the article is as follows:

1. Reply to Reviewer #1
   I have revised the article carefully

2. Reply to Reviewer #2
   The references for this article have been updated and discussed the limitations of this study.

3. Reply to EDITORIAL OFFICE’S COMMENTS
   I have modified the table as required and added new documents using the RCA tool.

2. The tables and references should be edited.
   I have changed the headings in Tables 1-5. References 1-18 were modified as requested by adding DOI and PMID. The fifth reference search failed to find DOI or PMID, so a link was included at the end of the reference.

3. The article has been linguistic polished.

4. The manuscript has been revised in accordance with the required format.
   We hope that therevision is acceptable and look forward to hearing from you soon.

With best wishes!